Author’s response to reviews

Title: The effect of glucocorticoid therapy on mortality in patients with rheumatoid arthritis and concomitant type II diabetes: a retrospective cohort study.

Authors:

Ruth Costello (ruth.costello@manchester.ac.uk)
Antonia Marsden (antonia.marsden@manchester.ac.uk)
Mohammad Movahedi (Mohammad.Movahedi@uhnresearch.ca)
Mark Lunt (mark.lunt@manchester.ac.uk)
Jenny Humphreys (jenny.humphreys@manchester.ac.uk)
Richard Emsley (richard.emsley@kcl.ac.uk)
William Dixon (will.dixon@manchester.ac.uk)

Version: 3 Date: 14 Nov 2019

Author’s response to reviews:

Dear Dr Athwal,

Thank you for your comment, we have responded below:

Assistant Editor Comments:

1. Authors’ Contributions

We have noted that the authors Mark Lunt and Jenny H Humphreys are missing in the listed authors' contributions. The individual contributions of ALL authors to the manuscript should be specified in the Authors’ Contributions section. Guidance and criteria for authorship can be found here:

http://www.biomedcentral.com/submissions/editorial-policies#authorship
Response: The authors contributions statement has been updated to: “REC contributed to study design, conducting the analyses, interpreting the data, drafting the manuscript, critically revising the manuscript, approving the final version and agrees to be accountable for all aspects of the work. AM, RE and WGD contributed to study design, interpreting the data, critically revising the manuscript, approving the final version and agrees to be accountable for all aspects of the work. MM contributed to acquiring the data, interpreting the data, critically revising the manuscript, approving the final version and agrees to be accountable for all aspects of the work. ML and JHH contributed to interpreting the data, critically revising the manuscript, approving the final version and agrees to be accountable for all aspects of the work.”

I look forward to hearing from you soon.

Yours Sincerely,

Professor William G Dixon